Author's response to reviews

Title: Predictors of early transition to palliative level of care among middle cerebral artery ischemic stroke patients and the influence of swallowing dysfunction: a retrospective cohort study

Authors:

CHRISTA O'HANA V SAN LUIS (christaohana14md@yahoo.com)
LOUISE D MCCULLOUGH (lmccullough@uchc.edu)
ILENE STAFF (istaff@arthosp.org)
GILBERT J FORTUNATO (gfortun@arthosp.org)

Version: 5 Date: 24 April 2013

Author's response to reviews: see over
April 24\textsuperscript{th}, 2013

MS: 1766774282889702

Dear Dr. Seymour,

We would like to express our gratitude for informing us regarding the acceptance of our manuscript “Dysphagia as a predictor of outcome and transition to palliative care among middle cerebral artery ischemic stroke patients.” Thank you again for spending time reviewing our manuscript and contributing significantly to this final version. I have uploaded an editable version of the final revised manuscript.

Again, thank you!

Sincerely,

Dr. San Luis and McCullough